Author's response to reviews

Title: Useful of a Thematic Network, on the Primary Healthcare: A controlled interventional study in a rural area.

Authors:

Maria Jesus Coma del Corral MD (mjcoma@uninet.edu)
Pedro Abaigar Luquin MD (paluquin@hgy.es)
Jose Cordero Guevara MD (jcordero@hgy.es)
Angel Olea Movilla MD (jolea@uninet.edu)
Gerardo Torres Torres MD (gtorres@uninet.edu)

Version: 2 Date: 18 July 2005

Author's response to reviews:

Dear editor:

First, I like to thank the reviewers for their work!

- We have made some language corrections, as the revisors suggested, and we hope it could be acceptable for publication

- We can't include p-values to determine if the different in results of the intervention and the control groups are statistically significant or not, because all results were measured in entire populations from both Health Centres, not in samples. However, in order to emphasize better the effect of the intervention and to compare the results obtained in both ZBH, we have calculated the diference of Relative Risk Reduction (RRR) in the rate, in the ZBH studied and ZBH Control.

- The intervention program include a comprehensive list of the different activities performed.

- The spanish population census and the direct method of standardization was used to adjust the rate of UTB by active population/pensionist. This explain is added in "Methods" in the manuscript

Thank you very much!

Maria Jesus Coma
Research Unit
Hospital "General Yague"
Burgos. Spain